

EO-1 Weekly Report


	Earth Observing – 1

	May 2, 2012 – May 8, 2012

	DOY
	123
	–
	129

	Mission Day
	4241
	–
	4247


EO-1 SPACECRAFT

Instruments

· Scheduled 124 science Data Collection Events (DCEs) this week 

· Lost a scene on DOY 123/12:23z lost Hyperion Instrument data for one scene on DOY 124/11:23z. 
· See the Discrepancies section bellow for more detail. 

· Performed instrument decontamination cycles

· Conducted HSI deicing from 128/00:25z to 128/15:15z

· Conducted ALI outgassing from 128/00:35z to 128/15:25z
· Performed Instrument calibration

· Solar Array Characterization at 128/10:36z

· HSI solar calibration 129/06:19z
EO-1 MISSION OPERATIONS CENTER

Real-Time

· Maintaining nominal spacecraft state of health

· Monitoring passes with the Trollsat ground station for receiver and commanding lock

· Monitoring passes with the PF1 ground station for connection issues

· Gathering a list of end of mission thruster commands 

Mission Planning
· Generating routing back-up ATS loads 
· Creating a PROC to support X-band passes with the Trollsat ground station

Flight Dynamics
· SGS tracking data is being correctly processed during orbit determination 

· Checked for locations on orbit of the SSPC anomalies
System Administration
· Completed configuring the 3rd string ASIST system as a standalone workstation. This system was previously NFS mounted to the primary ASIST system.

· Unmounted the NFS partition, edited the fstab so it would not mount it again, and pointed ASIST to the local directory structure instead of the NFS mount.

· Populated ASIST directories with data from the most recent primary ASIST backup.

· Disabled NFS on the primary ASIST computer.

· Continued work on the creation of an EO-1 central file server.

· Relocated systems into a rack in the MOC and connected to network and KVM.

· Updated the hosts table on all operational systems to include the new file servers.

· Updated EO-1 equipment list and sent an updated network diagram to IONet security

· Supported certification scan (no highs or mediums found) and received IONet approval to be on the network.

· Renewed expiration dates for 3 Open IONet firewall (OGRS) rules as they are still needed operationally.

· Completed annual CIS Benchmark review on another operational system. Verified previously applied benchmark settings and implemented new benchmark items. This review was documented in a benchmark spreadsheet.

· Completed the periodic reboot of the primary string of operational computers. A chkdsk/fsck is performed on each system during this routine maintenance (12 systems completed).

· Completed routine root/admin password changes on operational systems.

· Replaced failing monitor on the ASIST T&C system used by the Data Processing System.

· Attended code 400 training on the Symantec product that will be used for DAR (data at rest) encryption.

· Reported April POAM status update to IT Security team.

· Created backups for the ASIST computers, flight dynamics computers, ASPEN computers, countdown clock, data processing computers, FEDS computers, firewall, trending system, and MOC CNE computer.

· Created weekly tape.

GROUND AND SPACE NETWORK

Station Downtimes 

· SGS has had an equipment malfunction and latest indication is that it will be back up August 2012
· HGS is down, unknown time to be back up

Operational Discrepancies

· Loss of Scene DOY 123/12:23z
· X-band data files were received from the ground station, had good data for 280 seconds, ratty data for 36 seconds and then good data for 250 seconds

· Loss of Hyperion Data for one scene DOY 124/11:23z

·  X-band data files were received from the ground station, had good data for 190 seconds, ratty data for 50 seconds and then good data for 320 seconds
Number of Real-Time Supports

[image: image1.emf]Date DOY S/X-Band S-Band Only Non-DSMC TDRSS Daily Totals

2-May-12 123 5 6 1 0 12

3-May-12 124 6 5 0 0 11

4-May-12 125 6 3 0 0 9

5-May-12 126 6 3 1 0 10

6-May-12 127 6 3 0 0 9

7-May-12 128 2 10 0 0 12

8-May-12 129 6 3 0 0 9

37 33 2 0 72 Weekly Totals


UPCOMING EVENTS

· TSM update onboard spacecraft
· TR2 X-Band pass testing
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		Date		DOY		S/X-Band		S-Band Only		Non-DSMC		TDRSS		Daily Totals

		2-May-12		123		5		6		1		0		12

		3-May-12		124		6		5		0		0		11

		4-May-12		125		6		3		0		0		9

		5-May-12		126		6		3		1		0		10

		6-May-12		127		6		3		0		0		9

		7-May-12		128		2		10		0		0		12

		8-May-12		129		6		3		0		0		9

		Weekly Totals				37		33		2		0		72
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